
    PS Booster 

Machine Coordinator last week A.Akroh / F.Chapuis 

Machine Coordinator this week F.Chapuis / A.Akroh 

Beam Scheduled 

ISOLDE No PS No 

Beam Availability by Destination (AFT) 

ISOLDE None % PS None % 

Facility Status 

Summary Hardware-Test period continuation 

Issues 

Main issue (involving machine accesses and MPS lockout): 

• Water Cooled Cables (WCC) supplying the PSB main-magnets with loose 
fixings in BR.QFO132 → situation stabilized with clamps and checked 
during a second access on Friday 21/02/25. 
Equipment Specialists informs that all the fixings are fatigued and suggests 
changing them during LS3 (request forwarded to EN-ACE).  
 

Other minor issues solved on: BI.KSW, BTY.BTV. 
 
LIC (MTG) update done. 
 
Hardware-Test progression: 74% done as of end of Friday 21/02/25. 
 

Plans 

Hardware-Test continuation until Wednesday 26/02/25. 
Test schedule: 

• Finalization of the Interlock Systems tests (SIS_v1 and BIS/BIC) 

• Final Check → preparation of the beams in view of the next period: Beam 
Commissioning starting on Thursday 27/02/25.  
 
 

Intervention Request 

No Duration  Preferred date/time  

Reason  

Impact  

 


